Author's Response To Reviewer Comments

As per editorial instructions, we have:

- 1) Included citations to the GigaDB and ZENODO datasets wehre appropriate
- 2) Removed the sentence "All other data available upon request"
- 3) Added the MIT license to our github repo
- 4) Added a statement to the manuscript involving the approval for research with animals (Candian Wildlife Service permit and UToronto AUP).

2/5/18

5) Fixed the references 28, 29 and 38